Author's response to reviews

Title: Unusual association leiomyomatosis peritonealis disseminata with appendiceal endometriosis: a case report

Authors:

   Woo Yong Lee (yongaaa5972@naver.com)
   Ji Hyun Noh (tttnii@hanmail.net)

Version: 3 Date: 13 May 2015

Author's response to reviews: see over
May 13, 2015
Dear Editor

MS: 8187930531595900
Title: Unusual association leiomyomatosis peritonealis disseminata with appendiceal endometriosis: a case report
Author: Woo Yong Lee and Ji Hyun Noh

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated
2 Revision has been made according to the suggestions of the reviewer
   (1) Referee 1: We added and corrected the description of pathologic findings which is highlighted with yellow in main document.
   Additionally, I also modified and corrected the legend of figures.
   (2) Referee 2: We also additionally discussed references which Referee 2 mentioned.
   (3) We also ensured that our revised manuscript conforms to the journal style.

Thank you again for publishing our manuscript in the The Journal of Medical Case Reports
Sincerely yours,

Woo Yong Lee, MD,
Department of Surgery
Seoul Paik Hospital, Inje University College of Medicine,
85 Jeo-Dong 2-ga, Jung-Gu, Seoul, 100-032, Republic of Korea.
E-mail: yongaaa5972@naver.com
telephone: +82-2-2270-0247
fax: +82-2-2270-0250